Author's response to reviews

**Title:** Outcome of Retinopathy of Prematurity (ROP) Patients Following Adoption of Revised Indications for Treatment

**Authors:**

Aaron M Alme MD (aalme@unmc.edu)
Michael L Mulhern (mmulhern@unmc.edu)
Thomas W Hejkal MD, PhD (twhejkal@unmc.edu)
Jane L Meza PhD (jmeza@unmc.edu)
Fang Qiu MS (fqiu@unmc.edu)
David D Ingvoldstad MD (dingvold@gmail.com)
Eyal Margalit MD, PhD (emargalit@unmc.edu)

**Version:** 2  **Date:** 31 March 2008

**Author's response to reviews:**

Dear Dr. Marlee,

Thank you for the opportunity to revise our Abstract for peer review. We have reviewed the Abstract instructions and modified our Abstract to include only the 4 titles requested by BMC Ophthalmology: Background, Method, Results, Conclusions. It now has a more narrative flow. In addition, we added specific details of the statistical analysis under the Methods section, removed two abbreviations, and now believe that it is more readable to someone just being introduced to this area of medicine. We look forward to hearing from you; if there is any additional changes or modifications, do not hesitate to contact us.

Thank you,
Aaron Alme MD